Author's response to reviews

Title: The rise of assertive community interventions in South Africa: A randomized control trial assessing the impact of a modified assertive intervention on readmission rates; a three year follow-up

Authors:

Ulla A Botha (ulla@sun.ac.za)
Liezl Koen (liezl@sun.ac.za)
Ushma Galal (ullabotha@gmail.com)
Esme Jordaan (harriealex1@gmail.com)
Daniel JHN Niehaus (djhn@sun.ac.za)

Version: 4 Date: 16 January 2014

Author's response to reviews: see over
16th January 2014

The Editors
BMC Psychiatry

Sir/Madam

Submission:
The rise of assertive community interventions in South Africa: A randomized control trial assessing the impact of a modified assertive intervention on readmission rates; a three year follow-up

Botha U, Koen L, Galal U, Jordaan E, Niehaus DJHN

Thank you for giving us the opportunity to respond to the reviewers’ comments. We have found the feedback helpful and constructive and have attempted to answer all queries. We also appreciate the extension we have been granted to reply.

Please see the reply to both reviewers in the reviewer feedback report which was uploaded as a separate document.

Thanking you in advance

U Botha
Department of Psychiatry
University of Stellenbosch
PO Box 19063
Tygerberg 7505
South Africa
Fax: +27 21 938 9738
e-mail: ulla@sun.ac.za